An Accessible Web-Based Survey to Monitor the Mental Health of People With Mild Intellectual Disability or Low Literacy Skills During the COVID-19 Pandemic: Comparative Data Analysis

Monique CJ Koks-Leensen^{1,2*}, PhD; Anouk Menko^{1,3*}, MSc; Fieke Raaijmakers^{1,3,4}, MA, MSc; Gerdine AJ Fransen-Kuppens^{1,3,5}, Prof Dr; Kirsten E Bevelander^{1,2,3}, PhD

⁴Safety and Health Region Gelderland-Midden, Arnhem, Netherlands

⁵Municipal Health Service Gelderland Zuid, Nijmegen, Netherlands

*these authors contributed equally

Corresponding Author:

Monique CJ Koks-Leensen, PhD Department of Primary and Community Care Radboud university medical center Geert Grooteplein 21 Nijmegen, 6525 EZ Netherlands Phone: 31 243618181 Email: monique.koks-leensen@radboudumc.nl

Abstract

Background: The COVID-19 pandemic and related control measures affected the mental health of all populations. Particular subgroups are underrepresented in mainstream surveys because they are hard to reach, and study measurements are not adapted to their skills. These subgroups include people with lower cognitive and literacy skills, such as people with mild intellectual disability (MID), who were considered vulnerable during the COVID-19 pandemic given their low socioeconomic status, small social networks, increased risks of health problems, and difficulties understanding health-related information.

Objective: This study examines the impact of the COVID-19 pandemic on mental health among people with MID or low literacy skills compared with those predominantly represented in national surveys.

Methods: A repeated cross-sectional study of people with MID or low literacy skills and a general population sample was conducted in the Netherlands. An easy-read web-based survey was co-designed with, and tested among, people with MID or low literacy skills and conducted in 3 rounds within 1 year of the COVID-19 pandemic (T1: November to December 2020, T2: March to April 2021, and T3: September to October 2021). The survey contained questions about demographics and 6 aspects of mental health: feeling happy, feeling energized, feeling stressed, worry, feeling lonely, and sleeping problems.

Results: Our adapted survey and recruitment procedure enabled 1059 persons with MID or low literacy skills to participate (T1: n=412, 38.9%; T2: n=351, 33.1%; and T3: n=296, 28%). They were significantly younger, had a lower level of education, and more often than not were born outside the Netherlands compared to the general population sample (*P*<.001). Approximately half of them (604/1059, 57.03%) received professional care. They displayed poorer mental health scores than the general population sample. The percentages of people with MID or low literacy skills who reported more negative feelings in T1 ranged from 20.6% (85/412) reporting feeling lonely *often* or *almost always* to 57.8% (238/412) reporting feeling happy *almost never* or *sometimes*. The general population sample's percentages were 5.4% (160/2930) and 32.2% (941/2918), respectively. Although scores improved over time in both populations, the disproportional effects remained.

Conclusions: General COVID-19–related restrictions for the entire Dutch population affected people with MID or low literacy skills more negatively than the general population. Our study underscores the relevance of including these subpopulations in public health research because they are often overlooked in regular health data. An accessible web-based survey particularly targeted at this population enabled us to do so, and we reached a group of respondents significantly different from regular survey

¹Department of Primary and Community Care, Radboud university medical center, Nijmegen, Netherlands

²Academic Collaborative Intellectual Disability and Health - Sterker op Eigen Benen (SOEB), Nijmegen, Netherlands

³Academic Collaborative AMPHI - Integrated Health Policy, Nijmegen, Netherlands

participants. This survey's results provided insights into the health of people with MID or low literacy skills and gained knowledge to be used by care organizations and policy makers to reduce health disparities during a pandemic and in general.

(JMIR Public Health Surveill 2024;10:e44827) doi: 10.2196/44827

KEYWORDS

monitoring; mental health; intellectual disabilities; low literacy; COVID-19; web-based survey

Introduction

Studying COVID-19–Related Impact

The COVID-19 pandemic and related disease control measures affected the entire world. People were advised to adhere to strict hygiene measures and to work from home (if possible), and public places and nonessential shops were closed. In addition, social distancing and visiting restrictions were in place during lockdowns. In general, these restrictions had a major impact on daily routines, social contacts, and mental health [1-3], affecting some individuals more than others [4]. In the Netherlands, the Rijksinstituut voor Volksgezondheid en Milieu (RIVM; National Institute for Public Health and the Environment) started conducting a national longitudinal survey to monitor the impact of COVID-19 and related measures on Dutch citizens [5,6]. The monitor is informative regarding disease control strategies and policy making [7]; however, there is an external validity bias because participant characteristics show that a majority of the participants have a high educational level and are middle-aged or older and women [6,8,9]. Indeed, studies have demonstrated that particular subgroups are often underrepresented and excluded from mainstream surveys because general recruitment strategies are unsuitable for reaching them, and study measurements are not adapted to their cognitive level or literacy skills [10-12]. These groups often comprise people with lower socioeconomic status and higher risks of health problems, and, in the case of the COVID-19 pandemic, more difficulties complying with preventive measures [13-15], given their housing or work situation.

This study examines the impact of the COVID-19 pandemic and related restrictions on mental health and well-being among people with lower cognitive and literacy skills in addition to those who are predominantly represented in the national survey. An accessible survey based on the national survey was developed, and alternative recruitment techniques were used to specifically include these underrepresented subgroups.

Subpopulations at Risk for Greater COVID-19–Related Impact

In the Netherlands, approximately 19% of the adult population (ie, 2.5 million adults) have limited reading, writing, or numeracy skills [16]. These limitations have various causes, such as a low level of educational attainment, migrant background, parents' level of education and literacy, or low information-processing skills [17]. The last item plays an important role in people with mild intellectual disability (MID), who experience considerable limitations in both intellectual functioning and adaptive behavior and often need support in their daily life [18]. It is estimated that 4% to 8% of the Dutch population have an MID [19]. People with low literacy skills or MID often have limited work and income, poor health, and small social networks [20-24]. In general, studies have shown that people with low education and health literacy as well as those without social support, a stable income, a daily routine, and access to services are more at risk of mental health problems such as anxiety, general distress, and loneliness arising from the COVID-19 pandemic [1,3,5,14,25,26]. Therefore, it is likely that the COVID-19 pandemic had a higher impact on the mental health of people with MID or low literacy skills compared with the general population. However, during the first months of the pandemic, very limited knowledge was available about the impact on this subpopulation, and our study was set up to provide both these essential insights and practical recommendations for policy makers and care providers.

During the COVID-19 pandemic, studies on mental health specifically aimed at people with mild or more severe intellectual disability (ID) showed negative impacts as a result of social isolation or a lack of social support, the rapid changes in COVID-19-related measures and difficulty understanding these measures, difficulty accessing services, and disruption of daily routines [27-29]. Two European surveys among people with ID found that more than half reported stress or anxiety [30,31] or felt more anxious than usual because of the pandemic and subsequent lockdown [31]. A US survey found that 41% of the participants with ID had experienced more mental health problems or symptoms since the pandemic began; worry and stress were most often mentioned [32]. Similarly, people with low health literacy experienced more anxiety disorders, bouts of depression, and sleeping disorders during the COVID-19 pandemic than those showing sufficient health literacy [14]. Altogether, these studies-primarily conducted during the first lockdown periods-showed a great impact on the mental health of people with MID or low literacy skills, which contributed to an increase in preexisting inequalities in health and well-being [24,33].

The current underrepresentation in national surveillance and surveys of people with MID or low literacy skills, as well as the consequent lack of information about them, adds to existing health disparities. To better understand the impact of the COVID-19 pandemic on people with MID or low literacy skills, the specific factors driving this impact, and their specific needs, more information is urgently needed. Knowledge acquired through monitoring population health in its local context can provide a basis for government and health organizations to develop appropriate strategies to reduce this impact accordingly. In addition, the course of the pandemic and the ever-changing COVID-19-disease control strategies over time are important aspects regarding the context in which people were affected. Due to the rapidly changing situation and regulations during the pandemic (eg, when vaccinations were offered or restrictions

XSL•FO RenderX

were lifted), a dynamic impact on mental health was expected, and more insight is needed into how people responded to this unpredictable course.

Objectives

This study examines the impact of the COVID-19 pandemic on people with MID or low literacy skills in direct comparison with the general population and over the course of the pandemic at 3 different time points. A unique survey study was set up that complemented the RIVM national survey. This study developed an accessible version of the web-based survey together with representatives of the target population and used suitable techniques to reach people with MID or low literacy skills.

Methods

Study Design

A repeated cross-sectional study of people with MID or low literacy skills and a general population sample was conducted during the COVID-19 pandemic in the Netherlands. The inclusion criteria were people with reading and writing difficulties, aged ≥ 16 years, living in the Netherlands, and completion of the survey. No exclusion criteria were used. A control question to assess participants on literacy skills or intellectual ability was not included because this was expected to be too sensitive for the participants. For reasons of comparison, the same survey was presented to 2 panels used to represent the general Dutch population.

The survey was administered 3 times in a 1-year period between November 2020 and November 2021. The first survey (T1) was distributed during a nationwide second lockdown (November to December 2020), the second survey (T2) was administered after the second lockdown and when the Dutch vaccination program had started (April to May 2021), and the third survey (T3) was distributed after the summer when most COVID-19–related restrictions had been lifted (September to October 2021). Figure 1 shows the timeline and the severity of the COVID-19 pandemic in the Netherlands by means of excess mortality rates. Data were derived from Statistics Netherlands [34].

Figure 1. Timeline showing COVID-19–related measures, COVID-19 mortality cases, and the timing of survey periods. The number of excess mortalities indicates the severity of the COVID-19 waves.

Number of excess mortalities per month						
<1000	1000 - 3000	>3000				



Study Population and Recruitment

The surveys were disseminated via organizations working with people with MID or low literacy skills, such as advocacy organizations, care facilities for people with MID, language education organizations and libraries supporting and educating people with no or low literacy skills, social workplaces, the Dutch center of expertise on health disparities Pharos, Special Olympics, and a website offering accessible web-based information and programs for people with limited digital or literacy skills [35]. The surveys were open for between 4 and 6 weeks, giving the organizations time to distribute them within their network. Support was available to allow people who were

```
https://publichealth.jmir.org/2024/1/e44827
```

anxious or unable to complete the survey independently to participate in this study. At the end of each survey, participants were asked to participate in future research, resulting in a panel of participants who could be contacted directly for the subsequent survey rounds.

Our easy-read survey was also distributed within the same time period to 2 municipal health service (MHS) panels: GGD Gelderland-Zuid (n=approximately 2500) and GGD Gelderland-Midden (n=approximately 7000). Each MHS panel consisted of residents in its service area who are regularly asked to complete health surveys. There is a known bias to these panels, in that they generally consist of older and more highly

educated residents, with an overrepresentation of women [36]. We particularly used these panels in our study to obtain comparative data from a Dutch population sample and because these panels were also invited to participate in the national RIVM survey.

All participants received the same survey. Participation was voluntary, and participants could stop completing the survey at any time. Completion of a prior survey was not mandatory for participation in the next round. Data were obtained anonymously; therefore, matching between surveys and paired within-group analysis over time were not possible.

Web-Based Survey Development

The RIVM survey on Dutch citizens' perception of the COVID-19-related measures, their impact on well-being, and whether people were complying formed the basis of this monitor but was adapted to provide an easy-read version for this study [6]. The national survey consisted of >100 questions (eg, about participants' well-being, trust in the government, adherence to COVID-19-related measures, the risk of COVID-19 infection, and the understanding of COVID-19 information). Over time, new topics such as willingness to receive COVID-19 vaccinations and vaccine hesitancy were added to the survey. To create an easy-read version, we adapted the national survey in three steps by (1) shortening the survey (ie, we selected only a limited number of relevant topics), (2) reducing the number of response categories, and (3) adjusting the language level. This procedure was based on literature insights [37] and carried out in collaboration with professionals (researchers: n=2, care providers: n=3, and policy makers: n=2 working with people with MID or low literacy skills regarding health-related issues) as well as 2 experiential experts with MID or low literacy skills trained to advise, and experienced in advising, research projects. First, to shorten the survey, we discussed, prioritized, and selected the most relevant topics in light of our research objective to measure mental health, relevance to the target population, and the target population's cognitive capacity to answer the questions. Second, we reduced the number of response categories, in terms of Likert-scale options [38,39], by verifying the distinctiveness between the response categories and checking the relevance of the categories [40]. During the third and final step, abstract concepts, time references, and the language level were adjusted [41].

The easy-read survey consisted of 40 to 60 questions, with the length depending on the answers given to previous questions. The survey was pilot-tested with people with MID or low literacy skills (n=6) using the think-aloud methodology in cognitive interviews [42]. We used this method to verify the intended constructions of the questions and to assess the language level and the fit of the questions and response categories. Next, a web-based version of the survey was created on a web-based platform (called *I Coresearch*) designed with and for people with MID [43]. The platform has a clear layout, the possibility to enlarge font size, icons that can be added to response categories, and a speech-to-text and text-to-speech function. Additional pilot tests were carried out in which we observed participants (n=4-7 in each wave) while they were completing the survey to evaluate and improve the usability of

the platform. The observation sessions were followed by retrospective interviews. The tests resulted in minor adjustments to the questions, the response categories, and the web-based platform. This procedure resulted in a final short easy-read web-based survey.

These same steps were followed to modify and revise the survey for the second and third rounds. After each survey round, the findings were discussed in 4 to 5 focus groups with either people with MID or low literacy skills or care and support professionals and policy makers concerned with these subgroups. This not only led to a quick dissemination of our findings accompanied by solutions or practical tips to put into practice, but the group discussions also provided input for the subsequent survey rounds in which questions that became less relevant over time (eg, adherence to specific measures and difficulty coping with changes in specific daily activities) were replaced by new questions (eg, about vaccination).

Measures

The easy-read survey consisted of various topics, and we report the measures used for this specific study only (for details, refer to Multimedia Appendix 1).

Demographics and Contextual Factors

Similar to the national survey, an extensive section on demographics was included, such as age, gender, educational level, country of birth, and living situation. Furthermore, contextual factors known to have a potential influence on mental well-being were selected from the national survey and included, such as health status (eg, a rating of physical health and whether the participant had experienced COVID-19 infection), having social contacts, and socioeconomic status (eg, work status and the cessation of main activities because of COVID-19-related restrictions) [24,25]. To fit our target group's everyday experience, work status included paid work, volunteer work, school, and day care. In addition to the national survey questions, we included *receiving professional care* because this is an important characteristic describing the support needs of our target population, as well as survey completion methods (alone or with help), about which participants with MID or low literacy skills were asked (refer to Multimedia Appendix 1 for all questions listed in the easy-read questionnaire).

Mental Health

To gain a better understanding of the impact of the COVID-19 pandemic on mental health, a final set of 6 outcome measures regarding mental health were defined. The RIVM survey incorporated a mix of newly developed and existing validated scales or items in its well-being module, including psychological well-being (the 5-item Mental Health Inventory [44]), loneliness (the 6-item De Jong Gierveld Loneliness Scale [45]), life satisfaction, resilience, positive and negative effects experienced due to the COVID-19 pandemic, and emotional response (ie, the extent of worry, stress, or fear people experience) to monitor various aspects of mental health during the COVID-19 pandemic [6]. The most relevant items for our target group and research aim were selected from this set in the first developmental step of item generation. Subsequently, the response format was evaluated. For reasons of uniformity, the response categories

XSL•FO RenderX

in this series of questions were all adapted to a 4-point Likert scale. All questions had to be revised in accordance with the language level and understanding of the target group, and we ensured that overlapping concepts were avoided. These steps resulted in the following questions: (1) "Did you feel happy in the last couple of days?" (2) "Did you feel full of energy in the last couple of days?" (3) "Did you worry in the last couple of days?" (5) "Did you feel stressed in the last couple of days?" (5) "Did you feel lonely in the last couple of days?" and (6) "Did you have problems falling asleep in the last couple of days?" The outcomes were measured on a 4-point Likert scale: 1=yes, almost always; 2=yes, often; 3=yes, sometimes; and 4=no, almost never (Multimedia Appendix 1).

Statistical Analyses

Mental health was measured at 3 time points over a 1-year period among people with MID or low literacy skills (referred to as the target panel) as well as among members of the MHS panels. First, we calculated the frequencies and medians of the descriptive and contextual measures at each time point for the target panel and the MHS panels. To assess differences between the panels, in each round, Pearson chi-square tests were conducted for nominal or ordinal variables, and nonparametric t tests (2-tailed) were performed for age. Second, we calculated the differences in frequencies of mental health scores using Pearson chi-square tests between survey rounds within each panel and between panels for each survey round. Third, we analyzed the impact of the group differences on mental health using linear regression analyses, while controlling for genderand age-related differences. Given the large number of participants in each round and the multitude of comparisons made in analysis, differences and associations were considered statistically significant if P values were <.01 [46]. Statistical analyses were conducted in SPSS (version 25.0; IBM Corp).

Ethical Considerations

The study was reviewed by the medical research ethics committee of Radboud University Medical Center, which ruled that this study did not fall under the Medical Research Involving Human Subjects Act and was therefore exempt from formal ethical review (2020-7033). We conducted the study in accordance with the General Data Protection Regulation and standard operating procedures of our research center.

All participants received the survey after they had been fully informed, in plain language, about the purpose of this study. All participants provided web-based written informed consent regarding participation and the use of their data for this study and for future purposes before filling out each questionnaire. For each survey, 20 vouchers worth C0 (US \$53.9) each were raffled among people with MID or low literacy skills as motivation for participation.

Contact information used for the purpose of this raffle or future research was obtained and saved in a separate environment so that survey data could be obtained anonymously. Therefore, matching between surveys and paired within-group analysis over time were not possible.

Results

Participant Characteristics

Our web-based survey and adapted recruitment procedure enabled 1059 persons with MID or low literacy skills to participate (T1: n=412, 38.9%; T2: n=351, 33.1%; and T3: n=296, 28%). Background and contextual characteristics per survey round for the target panel and the MHS panels are presented in Table 1. Over the 3 time periods, 46.6% (138/296) to 53.2% (219/412) of the participants with MID or low literacy skills were women, with median ages ranging from 42 (IOR at T1: 27-57; IQR at T3: 28-54) to 45 (IQR 30-57 at T2) years and >70% (T1: 292/412, 70.9%; T2: 253/351, 72.1% and T3: 221/296, 74.7%) reporting no education or a low educational level. The majority (299/412, 72.6% at T1; 295/351, 84% at T2 and to 253/296, 85.5% at T3) were born in the Netherlands, 49.5% (204/412) to 62.2% (184/296) received professional care, and 22.6% (93/412) to 31.8% (94/296) reported living in a residential setting. Approximately half of the respondents (217/412, 52.7% at T1; 164/351, 46.7% at T2 and 170/296, 57.4% at T3) in each round reported very good or good physical health.

A total of 9305 MHS panel members completed our survey (T1: n=2930, 31.49%; T2: n=3213, 34.53%; and T3: n=3162, 33.98%). On the MHS panels over the 3 survey rounds, 55.85% (1766/3162) to 62.53% (1832/2930) of the participants were women, with median ages ranging from 52 to 62 years, and 12.73% (373/2930 at T1) to 17.08% (540/3162 at T3) had no education or a low educational level (ie, >70% had an intermediate or advanced educational level). The majority (2790/2930, 95.22% at T1; 3073/3213, 95.64% at T2 and 3023/3162, 95.6% at T3) were born in the Netherlands, only 3.28% (96/2930 at T1; 92/3213, 2.86% at T2 and 105/3162, 3.32% at T3) received professional care, and <1% (6/2930, 0.2% at T1; 6/3213, 0.19% at T2 and 3/3162, 0.1% at T3) lived in a residential setting. In addition, 77.12% (2478/3213 at T2 and 2435/3162, 77% at T3) to 79.39% (2326/2930 at T1) reported having very good or good physical health.

Altogether, this suggests that we successfully included a sample that represented our target population (ie, people with MID or low literacy skills). In addition, the characteristics of the MHS panels resemble those of the national sample, which is often used to represent the general Dutch population [9,36].



Koks-Leensen et al

Table 1. Distribution of demographic and contextual characteristics (health and socioeconomic status, social contacts, and target group–specific characteristics) per survey round by target panel and municipal health service (MHS) panels^a.

Characteristics	T1 ^b target panel (n=412)	T1 MHS panels (n=2930)	T2 ^c target panel (n=351)	T2 MHS panels (n=3213)	T3 ^d target panel (n=296)	T3 MHS panels (n=3162)
Age (y), median (IQR)	42 (27-57) ^e	52 (41-64)	45 (30-57) ^e	60 (45-71)	42 (28-54) ^e	62 (46-72)
Gender ^f (woman), n (%)	219 (53.2) ^g	1832 (62.5)	185 (52.7) ^h	1865 (58)	138 (46.6) ^e	1766 (55.9)
Educational level ^e , n (%)						
No to primary education	91 (22.1)	12 (0.4)	63 (17.9)	17 (0.5)	82 (27.7)	25 (0.8)
Low	201 (48.8)	361 (12.3)	190 (54.1)	477 (14.8)	139 (47)	515 (16.3)
Intermediate	53 (12.9)	769 (26.2)	64 (18.2)	811 (25.2)	45 (15.2)	790 (25)
Advanced	27 (6.6)	1730 (59)	8 (2.3)	1848 (57.5)	12 (4.1)	1785 (56.5)
Other ⁱ	39 (9.5)	56 (1.9)	25 (7.1)	59 (1.8)	17 (5.7)	46 (1.5)
Born in the Netherlands, n (%)	299 (72.6) ^e	2790 (95.2)	295 (84) ^e	3.073 (95.6)	253 (85.5) ^e	3.023 (95.6)
Living situation ^e , n (%)						
Alone	101 (24.5)	388 (13.2)	123 (35)	495 (15.4)	88 (29.7)	516 (16.3)
With family	209 (50.7)	2525 (86.2)	149 (42.5)	2708 (84.4)	112 (37.8)	2636 (83.4)
In residential setting	93 (22.6)	6 (0.2)	79 (22.5)	6 (0.2)	94 (31.8)	3 (0.1)
Very good or good physical health, n (%)	217 (52.7) ^e	2326 (79.4)	164 (46.7) ^e	2478 (77.1)	170 (57.4) ^e	2435 (77)
COVID-19 infection, n (%) ^j	_	_	61 (17.4) ^k	461 (14.3)	45 (15.2) ¹	364 (14.7)
Daily activities ^m , n (%)						
Paid work	145 (35.2) ^e	1.920 (65.5)	133 (37.9) ^e	1.628 (50.7)	117 (39.5) ⁿ	1.528 (48.3)
Volunteer work	92 (22.3) ^o	748 (25.5)	74 (21.1) ^p	652 (20.3)	62 (20.9) ^q	738 (23.3)
School	100 (24.3) ^e	154 (5.3)	42 (12) ^e	87 (2.7)	47 (15.9) ^e	73 (2.3)
Day care	121 (29.4) ^e	43 (1.5)	103 (29.3) ^e	32 (1)	103 (34.8) ^e	30 (0.9)
Other (sport, hobby, care, other)	197 (47.8) ^r	1558 (53.2)	179 (51) ^s	1760 (54.8)	176 (59.5) ^t	1668 (52.8)
None	32 (7.8) ^u	351 (12)	19 (5.4) ^e	379 (11.8)	9 (3) ^e	373 (11.8)
Change in daily activity ^v , n (%)						
Reduced or stopped	125 (30.3) ^e	385 (13.1)	98 (27.9) ^e	538 (16.7)	67 (22.6) ^e	229 (7.2)
Nothing to do (bored) ^j	_	_	18 (5.1) ^e	54 (1.7)	19 (6.4) ^e	49 (1.5)
Social contacts, n (%)						
No one I can talk to	42 (10.2) ^e	188 (6.4)	45 (12.8) ^e	134 (4.2)	$20(6.8)^{W}$	135 (4.3)
No one who can help me	59 (14.3) ^e	103 (3.5)	<i>34 (9.7)^e</i>	88 (2.7)	24 (8.1) ^e	79 (2.5)
Receives professional care, n (%)	204 (49.5) ^e	96 (3.3)	216 (61.5) ^e	92 (2.9)	184 (62.2) ^e	105 (3.3)
Survey completion, n (%)						
Alone	187 (45.4)	_	202 (57.5)	_	169 (57.1)	_
With help	223 (54.2)	_	147 (41.9)	_	127 (42.9)	_

^aCategory totals do not always add up to 100% because some categories (I don't know and I don't want to answer) and item nonresponse are not shown. Percentages are based on presented variable totals per category.

^bT1: first survey (November to December 2020).

^cT2: second survey (April to May 2021).

^dT3: third survey (September to October 2021).

https://publichealth.jmir.org/2024/1/e44827

^eValue for the target panel is significantly different from that for the regional panel (*P*<.001). Italicized values emphasize significance.

f<1% indicated their gender as "other."

^gValue for the target panel is significantly different from that for the regional panel (P=.001). Italicized values emphasize significance.

^hValue for the target panel is not significantly different from that for the regional panel (P=.15).

ⁱThe answer category "other" is chosen by respondents when it does not fit any of the provided options. This may be because they do not know their educational level, do not recognize their education category from the option list, or they were educated in a country other than the Netherlands. ^jThis question was added to the survey from T2.

^kValue for the target panel is not significantly different from that for the regional panel (*P*=.12).

¹Value for the target panel is not significantly different from that for the regional panel (P=.72).

^mRespondents could provide multiple answers; the category total can therefore add up to than 100%.

ⁿValue for the target panel is significantly different from that for the regional panel (*P*=.005). Italicized values emphasize significance.

^oValue for the target panel is not significantly different from that for the regional panel (*P*=.16).

^pValue for the target panel is not significantly different from that for the regional panel (*P*=.73).

^qValue for the target panel is not significantly different from that for the regional panel (P=.38).

^rValue for the target panel is not significantly different from that for the regional panel (P=.04).

^sValue for the target panel is not significantly different from that for the regional panel (P=.18).

^tValue for the target panel is not significantly different from that of the regional panel (P=.03).

^uValue for the target panel is not significantly different from that for the regional panel (P=.01).

^vThis variable is constructed concerning the daily activities of paid work, volunteer work, and day care.

^wValue for the target panel is not significantly different from that for the regional panel (*P*=.02).

Mental Health

The analyses of the distributions of the frequencies of mental health scores within the target panel show no differences between T1 and T2. There are significant differences between T1 and T3 regarding feeling happy, feeling energized, feeling stressed, worry, feeling lonely, and sleeping problems among people with MID or low literacy skills (Figure 2) and between T2 and T3 for these aspects, except for worry and feeling lonely; the percentage of people reporting positive feelings *often* or *almost always* increased, and the percentage of people reporting negative feelings *often* or *almost always* decreased over time. There were no differences observed regarding sleeping problems.

Regarding the MHS panels, there were significant differences between each survey round for feeling happy, feeling energized, and feeling stressed. For worry and feeling lonely, significant differences were observed only between T3 and the 2 previous rounds. The direction of the differences is similar to that observed in the target panels. Similar to the target panels, the MHS panels did not report differences regarding sleeping problems.

The analyses between the different panels within the survey rounds show that the percentage of participants in the target panel reporting negative feelings on mental health outcomes was significantly higher compared with the members of the MHS panels, especially within T1 and T2 (Figure 2); for example, looking at the 6 outcome measures within T1, the percentages of people who reported more negative feelings range from 20.8% (85/408) feeling lonely often or almost always to 58.3% (238/408) feeling happy almost never or only sometimes. The MHS panels show a different and more positive distribution on all outcome measures. The percentages of people on the MHS panels who reported more negative feelings range from 5.5% (160/2904) feeling lonely often or almost always to 32.2% (941/2918) feeling happy almost never or only sometimes. Figure 2 presents more details on the distribution of all mental health outcomes for the target panel and the MHS panels.



Koks-Leensen et al

Figure 2. Distribution of mental health outcomes for participants in the target panels and members of the municipal health service (MHS) panels at 3 different time periods. T1: first survey; T2: second survey; T3: third survey. ^aDifference between groups within each round (P<.001), ^bDifference between rounds within target group (T1 vs T3: P<.001), ^cDifference between rounds within panel (T1 vs T2, T2 vs T3, and T1 vs T3: P<.001), ^dDifference between rounds within target group (T1 vs T3: P=.002), ^eDifference between rounds within target group (T2 vs T3, and T1 vs T3: P=.001), ^fDifference between rounds within target group (T1 vs T3: P<.001), ^fDifference between rounds within target group (T1 vs T3: P<.001), ^fDifference between rounds within target group (T1 vs T3: P<.001), ^fDifference between rounds within target group (T1 vs T3: P<.001), ^fDifference between rounds within target group (T1 vs T3: P<.001), ^fDifference between rounds within target group (T1 vs T3: P<.001), ^fDifference between rounds within target group (T1 vs T3: P<.001), ^fDifference between rounds within target group (T1 vs T3: P<.001), ^fDifference between rounds within target group (T1 vs T3: P<.001), ^fDifference between rounds within target group (T1 vs T3: P<.001), ^hDifference between rounds within target group (T1 vs T3: P<.001), ^hDifference between rounds within target group (T1 vs T3: P<.001), ^hDifference between rounds within target group (T1 vs T3: P<.001), ^hDifference between rounds within target group (T1 vs T3: P<.001), ^hDifference between rounds within target group (T1 vs T3: P<.001), ^hDifference between rounds within target group (T1 vs T3: P<.001), ^hDifference between rounds within target group (T1 vs T3: P<.001), ^hDifference between rounds within target group (T1 vs T3: P<.001), ^hDifference between rounds within target group (T1 vs T3: P<.001), ^hDifference between rounds within target group (T1 vs T3: P<.001), ^hDifference between

Target group

MHS panel





Did you feel energized in the last couple of days?^a

■ Yes, almost always ■ Yes, often ■ Yes, sometimes ■ No, almost never



Did you worry in the last couple of days?^a

No, almost never Ves, sometimes Yes, often Yes, almost always



Did you feel stressed in the last couple of days?^a



Did you have problems falling asleep in the last couple of days?^a





Did you feel lonely in the last couple of days?^a

■ No, almost never ■ Yes, sometimes ■ Yes, often ■ Yes, almost always



Impact of Literacy Skills on Mental Health

Regression analyses adjusted for age and gender by using them as covariates show that the differences found between the 2

```
https://publichealth.jmir.org/2024/1/e44827
```

RenderX

panels exist for almost all mental health outcomes in each survey

round, except feeling stressed in T3 (P=.07). In addition, the

differences between the panels during T1 (age- and

gender-adjusted β ranging from -0.376 to 0.525) are larger than

those observed in T3 (age- and gender-adjusted β ranging from -0.257 to 0.509), except for sleeping problems. Table 2 presents

details for all time periods and outcome measures.

Table 2. Results for each survey round with panel as independent variable, mental health measures as dependent variables, and gender and age as covariates.

		T1 ^a		T2 ^b		T3 ^c	
		β (95% CI)	P value	β (95% CI)	P value	β (95% CI)	P value
Fee	eling happy	·					
	Panel	$-0.386^{\rm d}(-0.464 \ to \ -0.287)$	<.001	-0.385 (-0.481 to -0.288)	<.001	-0.257 (-0.362 to -0.152)	<.001
	Gender	-0.037 (-0.096 to 0.022)	.22	-0.092 (-0.151 to -0.034)	.002	-0.065 (-0.122 to -0.07)	.03
	Age	0.000 (-0.002 to 0.002)	.75	0.000 (-0.002 to 0.002)	.86	0.000 (-0.002 to 0.002)	.77
Feeling energized							
	Panel	-0.156 (-0.249 to -0.063)	.001	-0.243 (-0.340 to -0.145)	<.001	-0.143 (-0.249 to -0.037)	.008
	Gender	-0.039 (-0.102 to -0.023)	.22	-0.080 (-0.139 to -0.021)	.008	-0.058 (-0.117 to 0.000)	.05
	Age	0.004 (0.002 to 0.006)	<.001	0.003 (0.001 to 0.005)	.003	0.004 (0.002 to 0.006)	<.001
Wo	orry						
	Panel	0.328 (0.246 to 0.409)	<.001	0.324 (0.255 to 0.429)	<.001	0.227 (0.135 to 0.319)	<.001
	Gender	0.158 (0.104 to 0.213)	<.001	0.217 (0.165 to 0.270)	<.001	0.181 (0.131 to 0.232)	<.001
	Age	0.000 (-0.001 to 0.002)	.60	-0.002 (-0.004 to 0.000)	.02	-0.002 (-0.003 to 0.000)	.06
Feeling stressed							
	Panel	0.274 (0.195 to 0.353)	<.001	0.199 (0.117 to 0.281)	<.001	0.081 (-0.168 to 0.006)	.07
	Gender	0.158 (0.105 to 0.211)	<.001	0.189 (0.140 to 0.239)	<.001	0.195 (0.147 to 0.243)	<.001
	Age	-0.010 (-0.012 to -0.008)	<.001	-0.012 (-0.014 to -0.011)	<.001	-0.013 (-0.014 to -0.011)	<.001
Sle	eping prob	lems					
	Panel	0.525 (0.439 to 0.611)	<.001	0.518 (0.427 to 0.609)	<.001	0.509 (0.411 to 0.606)	<.001
	Gender	0.220 (0.163 to 0.278)	<.001	0.296 (0.241 to 0.351)	<.001	0.240 (0.187 to 0.294)	<.001
	Age	0.003 (0.001 to 0.005)	.002	0.000 (-0.002 to 0.002)	.90	0.002 (0.000 to 0.003)	.09
Fee	ling lonely						
	Panel	0.479 (0.409 to 0.549)	<.001	0.461 (0.387 to 0.535)	<.001	0.297 (0.229 to 0.365)	<.001
	Gender	0.095 (0.049 to 0.142)	<.001	0.161 (0.116 to 0.206)	<.001	0.071 (0.034 to 0.109)	<.001
	Age	-0.001 (-0.003 to 0.000)	.10	-0.002 (-0.004 to -0.001)	.002	-0.002 (-0.003 to 0.000)	.01

^aT1: first survey (November to December 2020).

^bT2: second survey (April to May 2021).

^cT3: third survey (September to October 2021).

^dItalicized values indicate significant regression results, with P<.01 level.

Discussion

Principal Findings

This is the first study to monitor the mental health and well-being of people with MID or low literacy skills and a general population sample over the course of 1 year during the COVID-19 pandemic. With our adapted web-based survey co-designed with representatives from our target population, we were able to reach subgroups that are usually underrepresented in surveys. Our study showed that feelings of happiness, energy, worry, stress, and loneliness improved in both populations over the course of the pandemic. However,

RenderX

the COVID-19 pandemic and related restrictions had a much bigger impact on the mental health of people with MID or low literacy skills than on that of the general population.

In general, our findings show that, during the second lockdown in the Netherlands (ie, at the time of the first survey round), people with MID or low literacy skills as well as the general population sample reported poorer mental well-being than 1 year later when all restrictions were lifted, and the COVID-19 infections became less severe (ie, at the time of the third survey round). These findings are in line with research on people in vulnerable positions [1] as well as the general population who experienced fewer negative feelings over the course of the COVID-19 pandemic [5]. Previous literature has shown that

ng strategies to maintain social in

Koks-Leensen et al

the impact on mental health and well-being is correlated to the stringency of disease control measures [25,47]; for example, the closure of social care services, workplaces, and day care activities negatively influenced daily structure and social interactions [29-31,48,49], thereby increasing stress and anxiety [50], and quarantines and social isolation were found to have an effect on loneliness, fear, and boredom [27,28,51]. When these measures were relaxed, there was a partial improvement in mental health [47]. Although our study was not designed to prove any causation between the stringency of disease control measures and mental health impact, our findings show a similar pattern of decreasing worries, stress, and loneliness whereas feelings of happiness and energy increased over time. Notably, this was also the case in our general population sample [1,3,5,14,25,26]; however, the pandemic disproportionately impacted people with MID or low literacy skills, who reported more negative mental health outcomes in all survey rounds.

Qualitative studies among people with MID show that long-term social restrictions in particular had an extensive impact on their daily life by limiting social connections and work activities [27,29]. Our target population reported these limitations in daytime activities to a greater extent than the general population sample. Interviews by Voermans et al [29] provide more in-depth assessment of the consequences of these limitations for people with MID, showing a major impact in terms of social isolation, difficulties coping with negative thoughts, struggles with autonomy in society, stigmatization, a lack of routine and purpose, boredom, and lower self-worth. As awareness is raised about the significant value of meaningful social contacts and daytime activities, professionals and policy makers should provide tailored policies that consider both health risks and the risks of social isolation. Societal participation initiatives should be organized and sustained for people with MID or low literacy skills, both during and outside of a pandemic.

Besides the disruptive impact of disease control measures on the target population's daily routines and social contacts, the high levels of confusion and uncertainty that resulted from the rapidly changing measures as well as fear and loss of control may have played a role in their reduced mental health [32] in periods of both stringent measures and relaxation of control measures [14]. In addition, people with lower health literacy skills are known to have less resilience, which affects their feelings of anxiety, stress, or worry [52,53], thereby putting them at greater risk of mental health problems. Ongoing support should be provided to enhance resources of resilience and coping strategies in people with MID or low literacy skills through either formal or informal caregivers.

Our findings highlight the need to prioritize the mental health consequences of the pandemic and the disease control measures for people with MID or low literacy skills [1,54,55]. The majority of our sample received support from formal and informal caregivers, who are an important source of support. Studies have shown detrimental effects on the mental health of these caregivers as well [28,56]. Therefore, we suggest tailoring generic disease control measures to the specific situations of groups considered vulnerable and their support system, instead of widely implementing measures such as social distancing, visiting restrictions, and closure of schools or day care facilities

https://publichealth.jmir.org/2024/1/e44827

(eg, by developing strategies to maintain social inclusion during pandemic challenges through a combination of supportive carers, assisted digital communication technologies, and safe social activities) [57,58]. Hence, engaging groups considered vulnerable and their support system in policy making and decision-making is essential in the tailoring process [59].

This study underscores the relevance of including people with MID or low literacy skills in health research and therefore endorses current calls to action in practice and science [60]. Health information systems are crucial for providing data for policy making and decision-making, but the underrepresentation of people with MID or low literacy skills in health data may lead to biased policy decisions, with adverse and detrimental effects on existing health disparities [13,33]. Previous research has suggested that, to reduce disparities and guide policy, researchers should evaluate how health outcomes are distributed among specific demographic groups and compare these distributions with those of the overall population [61], as was done in our study. Collecting information about people with MID or low literacy skills should become routine in demographic and public health data collection. We have shown that, by co-designing an adapted survey and using an accessible web-based platform and specific recruitment procedures, it is possible to collect information among people with MID or low literacy skills, even during lockdown periods; for example, the sample characteristics showed that participants with MID or low literacy skills differed from the general population in educational level, country of birth, and daily activities. With our adjusted approach to data collection, we were quickly able to obtain relevant information about people with MID or low literacy skills and disseminate our findings and recommendations, thereby facilitating policy makers to guide disease control measures and health promotion activities that address the immediate as well as longer-term health needs of people with MID or low literacy skills or other populations considered vulnerable.

Limitations of the Study

Executing a repeated cross-sectional survey among people with MID or low literacy skills during a pandemic is fraught with challenges. Therefore, our study has some limitations. First, we did not collect longitudinal data because we wanted to lower the threshold for participation by choosing an anonymous design. In addition, people were not obliged to complete all 3 surveys. As a result, it was impossible to track individual participants over time. Second, inevitably, the validated questions had to be revised to include people with MID or low literacy skills in our survey. However, we tested the questions in cognitive interviews, and the project team worked as inclusively as possible together with the target population to create a valid survey to obtain reliable data. More than half of the participants over all survey rounds (558/1059, 52.69%) were able to complete the web-based survey themselves, and support was arranged for the remaining group of respondents (497/1059, 46.93%). This should encourage future researchers to consider easy-read web-based surveys among people with MID or low literacy skills as long as target group representatives are closely involved in designing and testing these surveys. Third, our study started after the onset of the COVID-19 pandemic and lacks a

XSL•FO RenderX

baseline measurement of mental health before the pandemic. Therefore, it remains inconclusive as to whether people with MID or low literacy skills experienced greater mental health problems during the COVID-19 pandemic than before the pandemic. Fourth, we relied mostly on organizations in our network (eg, health care organizations and public libraries with special literacy programs) to contact and recruit people with MID or low literacy skills. Besides the possible sampling bias that this may have caused, we could not track how many people were approached to take part in the survey. Therefore, we were unable to report information about response rates. Fifth, because the survey was conducted on the web, those without access to the internet or sufficient digital literacy skills may have been excluded. Sixth and last, there was also a bias in our general population sample. However, although the sample was not representative of the Dutch general population in terms of age, gender, and educational level, it allowed us to contextualize our findings and gain a deeper understanding of the challenges faced by people with MID or low literacy skills compared with the general population sample, while controlling for differences in gender and age. We were able to do so because we used the same easy-read questionnaire in both groups to prevent survey bias. An open-ended question about the experience of members of the MHS panels with this type of questionnaire at T3 revealed that the majority of participants (1829/2337, 78.26%) appreciated this approach, given that a broader population was enabled to participate. Although this may indicate that easy-read questionnaires can be used for broader purposes and other populations, rather than being aimed specifically at people with

low literacy skills alone, this single question does not provide sufficient information regarding a broad survey approach, and more research is required.

Conclusions

In conclusion, our study enabled insight into the impact of the COVID-19 pandemic and related control measures on the mental health of people with MID or low literacy skills. General disease control measures for the entire Dutch population had a more negative impact on people with MID or low literacy skills than on the general population. Although mental health improved over the course of the pandemic in both populations as the COVID-19–related restrictions were gradually lifted, and the severity of the disease decreased over time, the disproportional effect remained. Professionals should be aware of this and pay attention to the needs of people with MID or low literacy skills in research, practice, and policy by tailoring measures that consider physical, social, and mental health effects and providing support to overcome such effects.

This study underscores the relevance of including people with MID or low literacy skills in public health research because they are often overlooked in regular health data. An accessible and structural web-based monitor for people with MID or low literacy skills enabled us to do so and provides better knowledge for care providers and policy makers to react to unexpected events such as a pandemic. To prevent existing health disparities from increasing, greater account should be taken of the impact of control measures on people who are relatively more vulnerable.

Acknowledgments

The authors offer grateful thanks to their coresearchers Anneke van der Cruijsen and Paméla Melkert who were involved during all steps of the study as well as to all *experts by experience* who participated in the development, testing, or evaluation of the web-based survey and supported survey completion or the discussion of the survey results. The authors thank all organizations that participated in the recruitment of participants and, in particular, MEE Gelderse Poort, GGD Gelderland-Zuid, GGD Gelderland-Midden, and Pharos for their valuable contributions and cooperation during the execution of the study. This study was supported by a grant from the Netherlands Organization for Health Research and Development (ZonMw; grant 1043002201000). The funder had no role in study design, data collection, data analysis, data interpretation, the writing of the manuscript, and the decision to publish.

Data Availability

The data sets generated and analyzed during this study are available in the Radboud Data Repository.

Authors' Contributions

MCJKL, FR, and GAJFK participated in the study conception and design. AM and MCJKL conducted the literature search and coordinated survey development and participant recruitment. KEB, AM, and MCJKL performed survey development, AM was involved in survey programming, and KEB facilitated the implementation of the survey on the web-based platform. AM, FR, and MCJKL contributed to the acquisition of data. AM accessed and verified the data; and KEB, AM, and MCJKL analyzed the data and plotted the tables and figures. AM and MCJKL drafted the manuscript. All authors participated in interpreting the data and study findings, critically reviewing and contributing to the revision of the manuscript, and approving the final version.

Conflicts of Interest

None declared.

Multimedia Appendix 1

Overview of all questions of the easy-read survey included in this study and response categories provided, separated into demographics and contextual measures, and in mental health items. [DOCX File, 21 KB-Multimedia Appendix 1]

References

- 1. Tackling the mental health impact of the COVID-19 crisis: an integrated, whole-of-society response. Organization for Economic Co-operation and Development. 2021. URL: https://www.oecd.org/coronavirus/policy-responses/ tackling-the-mental-health-impact-of-the-covid-19-crisis-an-integrated-whole-of-society-response-0ccafa0b/ [accessed 2024-04-05]
- 2. Glover RE, van Schalkwyk MC, Akl EA, Kristjannson E, Lotfi T, Petkovic J, et al. A framework for identifying and mitigating the equity harms of COVID-19 policy interventions. J Clin Epidemiol. Dec 2020;128:35-48. [FREE Full text] [doi: 10.1016/j.jclinepi.2020.06.004] [Medline: 32526461]
- Gloster AT, Lamnisos D, Lubenko J, Presti G, Squatrito V, Constantinou M, et al. Impact of COVID-19 pandemic on 3. mental health: an international study. PLoS One. 2020;15(12):e0244809. [FREE Full text] [doi: 10.1371/journal.pone.0244809] [Medline: 33382859]
- van der Ven LG, Duinhof EL, Dückers ML, Jambroes M, van Bon-Martens MJ. Conceptualizing Vulnerability for Health 4. Effects of the COVID-19 pandemic and the associated measures in Utrecht and Zeist: a concept map. Int J Environ Res Public Health. Nov 19, 2021;18(22):12163. [FREE Full text] [doi: 10.3390/ijerph182212163] [Medline: 34831915]
- 5. RIVM Behavioral Unit . Welbevinden en leefstijl tijdens de coronacrisis. National Institute for Public Health and the Environment. Utrecht. RIVM; 2022. URL: https://www.rivm.nl/gedragsonderzoek/maatregelen-welbevinden/ welbevinden-en-leefstijl
- 6. van den Boom W, van Dijk M, Snijders B, Luijben G, van der Laan J, Euser S, et al. Cohort profile: the Corona Behavioral Unit cohort, a longitudinal mixed-methods study on COVID-19-related behavior, well-being and policy support in the Netherlands. PLoS One. Jul 31, 2023;18(7):e0289294. [FREE Full text] [doi: 10.1371/journal.pone.0289294] [Medline: 37523360]
- RIVM Behavioral Unit. Applying behavioural science to COVID-19. National Institute for Public Health and the Environment. 7. Utrecht. RIVM; 2022. URL: https://www.rivm.nl/en/behavioural-science/ results-of-study-behavioural-measures-and-well-being
- 8. RIVM Behavioral Unit. Gedragwetenschappelijk onderzoek COVID-19: over dit onderzoek. National Institute for Public Health and the Environment. Utrecht. RIVM; 2023. URL: https://www.rivm.nl/gedragsonderzoek/maatregelen-welbevinden/ resultaten-21e-ronde/over-dit-onderzoek
- 9. Sanders JG, Spruijt P, van Dijk M, Elberse J, Lambooij MS, Kroese FM, et al. Understanding a national increase in COVID-19 vaccination intention, the Netherlands, November 2020-March 2021. Euro Surveill. Sep 2021;26(36):2100792. [FREE Full text] [doi: 10.2807/1560-7917.ES.2021.26.36.2100792] [Medline: 34505565]
- Rosencrans M, Tassé MJ, Kim M, Krahn GL, Bonardi A, Rabidoux P, et al. Ohio State University Nisonger RRTC on 10. HealthFunction. Invisible populations: Who is missing from research in intellectual disability? Res Dev Disabil. Dec 2021;119:104117. [doi: 10.1016/j.ridd.2021.104117] [Medline: 34736107]
- 11. Krahn GL. A call for better data on prevalence and health surveillance of people with intellectual and developmental disabilities. Intellect Dev Disabil. Oct 2019;57(5):357-375. [doi: 10.1352/1934-9556-57.5.357] [Medline: 31568742]
- Jahagirdar D, Kroll T, Ritchie K, Wyke S. Patient-reported outcome measures for chronic obstructive pulmonary disease 12. : the exclusion of people with low literacy skills and learning disabilities. Patient. Feb 16, 2013;6(1):11-21. [FREE Full text] [doi: 10.1007/s40271-013-0004-5] [Medline: 23417577]
- 13. Stormacq C, Van den Broucke S, Wosinski J. Does health literacy mediate the relationship between socioeconomic status and health disparities? Integrative review. Health Promot Int. Oct 01, 2019;34(5):e1-17. [doi: 10.1093/heapro/day062] [Medline: <u>30107564</u>]
- 14. Hermans L, Van den Broucke S, Gisle L, Demarest S, Charafeddine R. Mental health, compliance with measures and health prospects during the COVID-19 epidemic: the role of health literacy. BMC Public Health. Jul 10, 2021;21(1):1365. [FREE Full text] [doi: 10.1186/s12889-021-11437-w] [Medline: 34243741]
- Hawkins RB, Charles EJ, Mehaffey JH. Socio-economic status and COVID-19-related cases and fatalities. Public Health. 15. Dec 2020;189:129-134. [FREE Full text] [doi: 10.1016/j.puhe.2020.09.016] [Medline: 33227595]
- Algemene Rekenkamer. Aanpak van laaggeletterdheid. The Hague.; 2016. URL: https://www.rekenkamer.nl/publicaties/ 16. rapporten/2016/04/20/aanpak-van-laaggeletterdheid [accessed 2024-04-05]
- Grotlüschen A, Mallows D, Reder S, Sabatini J. Adults with Low Proficiency in Literacy or Numeracy. OECD Education 17. Working Papers. Paris. OECD Publishing; 2016. URL: https://www.oecd-ilibrary.org/docserver/5jm0v44bnmnx-en. pdf?expires=1713940220&id=id&accname=guest&checksum=A4CB65144AB89FB2F111EB6CA7EF5F06 [accessed] 2024-05-07]

- Schalock R, Luckasson R, Tassé MJ. An overview of intellectual disability: definition, diagnosis, classification, and systems of supports (12th ed.). Am J Intellect Dev Disabil. Nov 01, 2021;126(6):439-442. [doi: <u>10.1352/1944-7558-126.6.439</u>] [Medline: <u>34700345</u>]
- 19. Woittiez I, Eggink EM. Het aantal mensen met een licht verstandelijke beperking: een schatting. Sociaal en Cultureel Planbureau. The Hague.; 2019. URL: <u>https://www.scp.nl/publicaties/publicaties/2019/10/01/</u>
- <u>het-aantal-mensen-met-een-licht-verstandelijke-beperking-een-schatting</u> [accessed 2024-04-05]
 20. Emerson E, Parish S. Intellectual disability and poverty: introduction to the special section. J Intellect Dev Disabil. Dec 30, 2010;35(4):221-223. [doi: 10.3109/13668250.2010.525869] [Medline: 21117879]</u>
- Europe Monitor: How low literacy impacts us all. Macroeconomic update Europe. PwC. 2018. URL: <u>https://www.pwc.nl/</u> nl/themas/europe-monitor/documents/pwc-europe-monitor-april-2018.pdf [accessed 2024-05-07]
- 22. van der Heide I, Rademakers J. Laaggeletterdheid en gezondheid: stand van zaken. Utrecht. Nivel; 2015. URL: <u>https://www.nivel.nl/nl/publicatie/laaggeletterdheid-en-gezondheid-stand-van-zaken</u> [accessed 2024-04-05]
- 23. van der Velden PG, Contino C, Das M, van Loon P, Bosmans MW. Anxiety and depression symptoms, and lack of emotional support among the general population before and during the COVID-19 pandemic. A prospective national study on prevalence and risk factors. J Affect Disord. Dec 01, 2020;277:540-548. [FREE Full text] [doi: 10.1016/j.jad.2020.08.026] [Medline: 32889378]
- 24. van der Heide I, Wang J, Droomers M, Spreeuwenberg P, Rademakers J, Uiters E. The relationship between health, education, and health literacy: results from the Dutch adult literacy and life skills survey. J Health Commun. 2013;18 Suppl 1(Suppl 1):172-184. [FREE Full text] [doi: 10.1080/10810730.2013.825668] [Medline: 24093354]
- 25. Scholz N. Mental health and the pandemic. European Parliamentary Research Service. 2021. URL: <u>http://www.europarl.europa.eu/RegData/etudes/BRIE/2021/696164/EPRS_BRI(2021)696164_EN.pdf</u>
- 26. COVID-19: protecting people and societies. Organization for Economic Co-operation and Development. 2020. URL: <u>https://www.oecd.org/coronavirus/policy-responses/covid-19-protecting-people-and-societies-e5c9de1a/</u> [accessed 2024-04-05]
- Embregts PJ, van den Bogaard KJ, Frielink N, Voermans MA, Thalen M, Jahoda A. A thematic analysis into the experiences of people with a mild intellectual disability during the COVID-19 lockdown period. Int J Dev Disabil. Oct 05, 2022;68(4):578-582. [FREE Full text] [doi: 10.1080/20473869.2020.1827214] [Medline: 35937180]
- Fisher MH, Sung C, Kammes RR, Okyere C, Park J. Social support as a mediator of stress and life satisfaction for people with intellectual or developmental disabilities during the COVID-19 pandemic. J Appl Res Intellect Disabil. Jan 11, 2022;35(1):243-251. [FREE Full text] [doi: 10.1111/jar.12943] [Medline: 34633129]
- 29. Voermans MA, den Boer MC, Wilthagen T, Embregts PJ. Long-term social restrictions and lack of work activities during the COVID-19 pandemic: impact on the daily lives of people with intellectual disabilities. Disabil Rehabil. Dec 18, 2023;45(24):4122-4132. [doi: 10.1080/09638288.2022.2147227] [Medline: 36398471]
- McCarron M, McCausland D, Luus R, Allen A, Sheerin F, Burke E, et al. The impact of coronavirus disease 2019 (COVID-19) on older adults with an intellectual disability during the first wave of the pandemic in Ireland. HRB Open Res. 2021;4:93. [FREE Full text] [doi: 10.12688/hrbopenres.13238.2] [Medline: 35112049]
- Amor AM, Navas P, Verdugo MÁ, Crespo M. Perceptions of people with intellectual and developmental disabilities about COVID-19 in Spain: a cross-sectional study. J Intellect Disabil Res. May 08, 2021;65(5):381-396. [FREE Full text] [doi: 10.1111/jir.12821] [Medline: 33555099]
- Rosencrans M, Arango P, Sabat C, Buck A, Brown C, Tenorio M, et al. The impact of the COVID-19 pandemic on the health, wellbeing, and access to services of people with intellectual and developmental disabilities. Res Dev Disabil. Jul 2021;114:103985. [FREE Full text] [doi: 10.1016/j.ridd.2021.103985] [Medline: 34049229]
- 33. Krahn GL, Hammond L, Turner A. A cascade of disparities: health and health care access for people with intellectual disabilities. Ment Retard Dev Disabil Res Rev. Jan 24, 2006;12(1):70-82. [doi: 10.1002/mrdd.20098] [Medline: 16435327]
- 34. Gezondheid in coronatijd. Centraal Bureau voor de Statistiek. URL: <u>https://www.cbs.nl/nl-nl/visualisaties/</u> welvaart-in-coronatijd/gezondheid-in-coronatijd [accessed 2024-04-29]
- 35. Steffie. URL: https://www.steffie.nl/ [accessed 2024-04-29]
- 36. Tabellenboek gezondheidsmonitor volwassenen regio gelderland-zuid. GGD Gelderland-Zuid. 2021. URL: <u>https://gezondheidincijfers.ggdgelderlandzuid.nl/handlers/ballroom.ashx?function=download&id=99</u> [accessed 2024-05-07]
- 37. Kooijmans R, Mercera G, Langdon PE, Moonen X. The adaptation of self-report measures to the needs of people with intellectual disabilities: A systematic review. Clin Psychol Sci Pract. Sep 2022;29(3):250-271. [doi: 10.1037/cps0000058]
- Rivera-Riquelme M, Piqueras JA, Cuijpers P. The revised Mental Health Inventory-5 (MHI-5) as an ultra-brief screening measure of bidimensional mental health in children and adolescents. Psychiatry Res. Apr 2019;274:247-253. [doi: 10.1016/j.psychres.2019.02.045] [Medline: 30818147]
- Fang J, Fleck MP, Green A, McVilly K, Hao Y, Tan W, et al. The response scale for the intellectual disability module of the WHOQOL: 5-point or 3-point? J Intellect Disabil Res. Jun 25, 2011;55(6):537-549. [doi: 10.1111/j.1365-2788.2011.01401.x] [Medline: 21435065]
- 40. Hartley SL, MacLean Jr WE. A review of the reliability and validity of likert-type scales for people with intellectual disability. J Intellect Disabil Res. Nov 25, 2006;50(Pt 11):813-827. [doi: 10.1111/j.1365-2788.2006.00844.x] [Medline: 16999781]

- 41. Finlay WM, Lyons E. Methodological issues in interviewing and using self-report questionnaires with people with mental retardation. Psychol Assess. Sep 2001;13(3):319-335. [doi: 10.1037//1040-3590.13.3.319] [Medline: 11556269]
- 42. Wolcott MD, Lobczowski NG. Using cognitive interviews and think-aloud protocols to understand thought processes. Curr Pharm Teach Learn. Feb 2021;13(2):181-188. [doi: 10.1016/j.cptl.2020.09.005] [Medline: <u>33454077</u>]
- 43. Ik onderzoek mee. URL: <u>https://www.ikonderzoekmee.nl/</u> [accessed 2024-04-29]
- 44. Berwick DM, Murphy JM, Goldman PA, Ware Jr JE, Barsky AJ, Weinstein MC. Performance of a five-item mental health screening test. Med Care. Feb 1991;29(2):169-176. [doi: 10.1097/00005650-199102000-00008] [Medline: 1994148]
- 45. Gierveld JD, Tilburg TV. A 6-item scale for overall, emotional, and social loneliness: confirmatory tests on survey data. Res Aging. Aug 18, 2016;28(5):582-598. [doi: 10.1177/0164027506289723]
- 46. Benjamin DJ, Berger JO, Johannesson M, Nosek BA, Wagenmakers EJ, Berk R, et al. Redefine statistical significance. Nat Hum Behav. Jan 1, 2018;2(1):6-10. [FREE Full text] [doi: 10.1038/s41562-017-0189-z] [Medline: 30980045]
- 47. RIVM Behavioral Unit. Gedrag, Welzijn en vertrouwen tijdens de COVID-19 pandemie: trends, verklaringen en geleerde lessen. National Institute for Public Health and the Environment, Netherlands. 2023. URL: <u>https://www.rivm.nl/documenten/gedrag-welzijn-en-vertrouwen-tijdens-covid-19-pandemie-trends-verklaringen-en-geleerde</u> [accessed 2024-05-07]
- Scheffers F, van Vugt E, Moonen X. Resilience in the face of adversity in adults with an intellectual disability: a literature review. J Appl Res Intellect Disabil. Sep 09, 2020;33(5):828-838. [FREE Full text] [doi: 10.1111/jar.12720] [Medline: 32153087]
- 49. Courtenay K, Perera B. COVID-19 and people with intellectual disability: impacts of a pandemic. Ir J Psychol Med. Sep 2020;37(3):231-236. [FREE Full text] [doi: 10.1017/ipm.2020.45] [Medline: 32404232]
- Navas P, Amor AM, Crespo M, Wolowiec Z, Verdugo MÁ. Supports for people with intellectual and developmental disabilities during the COVID-19 pandemic from their own perspective. Res Dev Disabil. Jan 2021;108:103813. [FREE Full text] [doi: 10.1016/j.ridd.2020.103813] [Medline: 33271448]
- Brooks SK, Webster RK, Smith LE, Woodland L, Wessely S, Greenberg N, et al. The psychological impact of quarantine and how to reduce it: rapid review of the evidence. Lancet. Mar 14, 2020;395(10227):912-920. [FREE Full text] [doi: 10.1016/S0140-6736(20)30460-8] [Medline: 32112714]
- Xiao X, Xiao J, Yao J, Chen Y, Saligan L, Reynolds NR, et al. The role of resilience and gender in relation to infectious-disease-specific health literacy and anxiety during the COVID-19 pandemic. Neuropsychiatr Dis Treat. 2020;16:3011-3021. [FREE Full text] [doi: 10.2147/NDT.S277231] [Medline: 33324062]
- Zlotnick C, Dryjanska L, Suckerman S. Health literacy, resilience and perceived stress of migrants in Israel during the COVID-19 pandemic. Psychol Health. Sep 12, 2022;37(9):1076-1092. [doi: <u>10.1080/08870446.2021.1921177</u>] [Medline: <u>33980091</u>]
- Torales J, O'Higgins M, Castaldelli-Maia JM, Ventriglio A. The outbreak of COVID-19 coronavirus and its impact on global mental health. Int J Soc Psychiatry. Jun 31, 2020;66(4):317-320. [doi: <u>10.1177/0020764020915212</u>] [Medline: <u>32233719</u>]
- 55. Latoo J, Haddad PM, Mistry M, Wadoo O, Islam SM, Jan F, et al. The COVID-19 pandemic: an opportunity to make mental health a higher public health priority. BJPsych Open. Sep 20, 2021;7(5):e172. [FREE Full text] [doi: 10.1192/bjo.2021.1002] [Medline: 34538286]
- 56. Zonneveld E, van Schelven F, Boeije H. Effects of the COVID-19 pandemic on quality of life among relatives of individuals with intellectual disabilities: a longitudinal study. J Appl Res Intellect Disabil. Jan 22, 2023;36(1):68-77. [FREE Full text] [doi: 10.1111/jar.13035] [Medline: 36134473]
- 57. Tromans S, Kinney M, Chester V, Alexander R, Roy A, Sander JW, et al. Priority concerns for people with intellectual and developmental disabilities during the COVID-19 pandemic. BJPsych Open. Oct 29, 2020;6(6):e128. [FREE Full text] [doi: 10.1192/bjo.2020.122] [Medline: 33118913]
- 58. Lunsky Y, Jahoda A, Navas P, Campanella S, Havercamp SM. The mental health and well-being of adults with intellectual disability during the COVID-19 pandemic: a narrative review. J Policy Pract Intellect Disabil. Mar 25, 2022;19(1):35-47. [FREE Full text] [doi: 10.1111/jppi.12412] [Medline: 35601014]
- 59. Stadnick NA, Cain KL, Oswald W, Watson P, Ibarra M, Lagoc R, et al. Co-creating a theory of change to advance COVID-19 testing and vaccine uptake in underserved communities. Health Serv Res. Jun 04, 2022;57 Suppl 1(Suppl 1):149-157. [FREE Full text] [doi: 10.1111/1475-6773.13910] [Medline: 35243622]
- 60. Landes SD, Turk MA. Health equity for people with intellectual and developmental disability requires vast improvements to data collection: lessons from the COVID-19 pandemic. Disabil Health J. Jan 2024;17(1):101539. [doi: 10.1016/j.dhjo.2023.101539] [Medline: <u>37783651</u>]
- Almeida C, Braveman P, Gold MR, Szwarcwald CL, Ribeiro JM, Miglionico A, et al. Methodological concerns and recommendations on policy consequences of the World Health Report 2000. Lancet. May 26, 2001;357(9269):1692-1697. [doi: <u>10.1016/s0140-6736(00)04825-x</u>] [Medline: <u>11425394</u>]

Abbreviations

ID: intellectual disability



MHS: municipal health serviceMID: mild intellectual disabilityRIVM: Rijksinstituut voor Volksgezondheid en Milieu (National Institute for Public Health and the Environment)

Edited by A Mavragani; submitted 23.12.22; peer-reviewed by J Gamble, S Gordon; comments to author 08.12.23; revised version received 30.01.24; accepted 20.03.24; published 30.05.24

<u>Please cite as:</u> Koks-Leensen MCJ, Menko A, Raaijmakers F, Fransen-Kuppens GAJ, Bevelander KE An Accessible Web-Based Survey to Monitor the Mental Health of People With Mild Intellectual Disability or Low Literacy Skills During the COVID-19 Pandemic: Comparative Data Analysis JMIR Public Health Surveill 2024;10:e44827 URL: <u>https://publichealth.jmir.org/2024/1/e44827</u> doi: <u>10.2196/44827</u> PMID: <u>38607229</u>

©Monique CJ Koks-Leensen, Anouk Menko, Fieke Raaijmakers, Gerdine AJ Fransen-Kuppens, Kirsten E Bevelander. Originally published in JMIR Public Health and Surveillance (https://publichealth.jmir.org), 30.05.2024. This is an open-access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work, first published in JMIR Public Health and Surveillance, is properly cited. The complete bibliographic information, a link to the original publication on https://publichealth.jmir.org, as well as this copyright and license information must be included.

